Reviewer #1:
Scientific Quality: Grade C (Good)
Language Quality: Grade B (Minor language polishing)
Conclusion: Minor revision
Specific Comments to Authors: A concise, comprehensive, and well structured case report. I think this paper is welcome and relevant for medical daily practice. I suggest you delete the previous corrected text in order to be more easy to read. Thank you.

Answer: I have deleted the previous corrected text.

Reviewer #2:
Scientific Quality: Grade C (Good)
Language Quality: Grade B (Minor language polishing)
Conclusion: Minor revision
Specific Comments to Authors: Hi thank you for inviting me to review the case report. the discussion should be improved. please describe the initial sign, symptoms and also physical examination.

Answer:
The patient was first found to be ill because of an increased blood count during a physical examination, with no discomfort or signs.

Physical examination
On physical examination, her vital signs were as follows: Body temperature, 36.3°C; blood pressure, 118/70 mmHg; heart rate, 82 bpm; respiratory rate, 19 breaths/min. She was pale and had signs of edema in both lower limbs on examination.

Reviewer #3:
Scientific Quality: Grade B (Very good)
Language Quality: Grade B (Minor language polishing)
Conclusion: Accept (High priority)
Specific Comments to Authors: The manuscript is well written regarding the rare case of light chain amyloidosis in a patient with internodal marginal zone lymphoma and Waldenstrom macroglobulinemia. The manuscript also describes systemic treatment and is written in good language and clarity. Writing this case where clinical presentation
lacks specificity and hence requiring detailed clinical evaluation and active search for the cause to avoid missed diagnosis and misdiagnosis along with regular follow-up and timely treatment to prolong survival are important take home messages from the report. It would be better if you could clarify the patient's presenting complaint. What was bone marrow biopsy reports in regards to Congo red stain?

Answer:

Chief complaints

A 65-year-old woman was admitted to a local hospital due to increased blood sedimentation during physical examination 6 years ago.

The bone marrow biopsy reports in regards to Congo red was negative. I added the result in the article.